Author's response to reviews

Title: Characterization of the Inflammatory Cell Infiltrate and Expression of Costimulatory Molecules in Chronic Echinococcus Granulosus Infection of the Human Liver

Authors:

Ali Vatankhah (avatankhah@hotmail.com)
Judit Halász (halaszijudi@gmail.com)
Violetta Piurko (piurko.violetta@gmail.com)
Tamás Barbai (tbarbai@gmail.com)
Erzsébet Rásó (rasoerzs@gmail.com)
Prof Jozsef Timar (jtimar@gmail.com)

Version: 3  Date: 3 September 2015

Author's response to reviews:

We are delighted that the reviewers accepted the results of our study. The revision was made according to the suggestion of the two reviewers. Rev1 asked stylistic corrections. We have corrected the English and style of the manuscript. Rev2 suggested to complete the Introduction part with cytokine profile data of EG infected patients. In lanes 88-90 one can find reference to those papers (ref15-17). Rev2. also suggested to discuss the role for Kupffer cells which is incorporated into the Discussion section, lane 334,ref47. Rev2. also asked for a more clear structured Abstract. We have rewritten the abstract, and made it more concise and clear. Hope with these changes our paper would be publishable in Your prestigious Journal.